Optimal pharmacotherapy pathway in adults with diabetic peripheral neuropathic pain: the OPTION-DM RCT

Solomon Tesfaye, 1,2* Gordon Sloan, 1
Jennifer Petrie, 3 David White, 3 Mike Bradburn, 3
Tracey Young, 4 Satyan Rajbhandari, 5
Sanjeev Sharma, 6 Gerry Rayman, 6 Ravikanth Gouni, 7
Uazman Alam, 8,9 Steven A Julious, 10 Cindy Cooper, 3
Amanda Loban, 3 Katie Sutherland, 3 Rachel Glover, 3
Simon Waterhouse, 3 Emily Turton, 3
Michelle Horspool, 11 Rajiv Gandhi, 1
Deirdre Maguire, 12 Edward Jude, 13,14
Syed Haris Ahmed, 8,15 Prashanth Vas, 16
Christian Hariman, 17 Claire McDougall, 18
Marion Devers, 19 Vasileios Tsatlidis, 20
Martin Johnson, 21 Didier Bouhassira, 22
David L Bennett 23 and Dinesh Selvarajah 2
on behalf of the OPTION-DM group

¹Sheffield Teaching Hospitals NHS Foundation Trust, Sheffield, UK

²Department of Oncology and Human Metabolism, Medical School, University of Sheffield, Sheffield, UK

³Clinical Trials Research Unit, University of Sheffield, School of Health and Related Research (ScHARR), Sheffield, UK

⁴School of Health and Related Research (ScHARR), University of Sheffield, Sheffield, UK

⁵Lancashire Teaching Hospitals NHS Trust, Chorley, UK

⁶East Suffolk and North Essex NHS Foundation Trust, Ipswich, UK

⁷Nottingham University Hospitals NHS Trust, Nottingham, UK

⁸University of Liverpool, Liverpool, UK

⁹Liverpool University Hospital NHS Foundation Trust, Liverpool, UK

¹⁰Medical Statistics Group, School of Health and Related Research (ScHARR), University of Sheffield, Sheffield, UK

¹¹NHS Sheffield Clinical Commissioning Group, Sheffield, UK

¹²Harrogate and District NHS Foundation Trust, Harrogate, UK

¹³Tameside and Glossop Integrated Care NHS Foundation Trust, Ashton under Lyne, UK

¹⁴University of Manchester, Manchester, UK

- ¹⁵Countess of Chester Hospital NHS Foundation Trust, Chester, UK
- ¹⁶King's College Hospital NHS Foundation Trust, London, UK
- ¹⁷Royal Wolverhampton NHS Trust, Wolverhampton, UK
- ¹⁸University Hospital Hairmyres, East Kilbride, UK
- ¹⁹University Hospital Monklands, Airdrie, UK
- ²⁰Gateshead Health NHS Foundation Trust, Gateshead, UK
- ²¹hVIVO Services Limited, London, UK
- ²²Hospital Ambroise Paré, Paris, France
- ²³Nuffield Department of Clinical Neurosciences, University of Oxford, Oxford, UK

Declared competing interests of authors: Solomon Tesfaye reports honoraria as speaker fees from Wörwag Pharma (Stuttgart, Germany), Pfizer Inc. (Pfizer Inc., New York, NY, USA), Novo Nordisk (Bagsværd, Denmark), Merck & Co. Inc. (Kenilworth, NJ, USA), Eva Pharma (Cairo, Egypt), Hikma Pharmaceuticals plc (London, UK), Abbott Laboratories (Abbott Park, IL, USA), AstraZeneca (Cambridge, UK), Nevro (Redwood City, CA, USA), Procter & Gamble Health Limited (Mumbai, India), Astellas Pharma Inc. (Tokyo, Japan) and Berlin-Chemi (Berlin, Germany), and is on the advisory boards (from 2018 to present) of Bayer AG (Leverkusen, Germany), NeuroPn Therapeutics (Norcross, GA, USA), Wörwag Pharma, Angelini (Rome, Italy), Grünenthal (Aachen, Germany), TRIGOcare International GmbH (Wiehl, Germany), Nevro, Mitsubishi Tanabe Pharma Corporation (Osaka, Japan) and Confo Therapeutics (Gent, Belgium). Uazman Alam reports honoraria for educational meetings from Eli Lilly and Company (Indianapolis, IN, USA), Napp Pharmaceuticals Ltd (Cambridge, UK), Sanofi (Paris, France) and Boehringer Ingelheim (Ingelheim am Rhein, Germany). Edward Jude reports honoraria and research support from AstraZeneca, Bayer AG, Menarini (Florence, Italy), Napp Pharmaceuticals Ltd, Novo Nordisk and Sanofi. Syed Haris Ahmed reports honoraria for educational meetings from Novo Nordisk (Bagsværd, Denmark), Eli Lilly and Company (Indianapolis, IN, USA) and Sanofi (Paris, France). Prashanth Vas reports honoraria from Merck & Co. Inc. and Sanofi. Martin Johnson reports honoraria for advisory boards and speaker fees from Grünenthal (2015 to present), and is a co-chairperson, since 2014, of the Chronic Pain Policy Coalition (London, UK) and a council (2012 to present) and ordinary member of the British Pain Society (London, UK). Didier Bouhassira reports honoraria for consulting activities for Bayer AG, Grünenthal, Novartis Pharmaceuticals UK Ltd (London, UK) and Air Liquide (Paris, France). David L Bennett has acted as a consultant on behalf of Oxford University Innovation Limited (Oxford, UK) for AditumBio (San Francisco, CA, USA), Amgen Inc. (Thousand Oaks, CA, USA), Bristows LLP (London, UK), Latigo Biotherapeutics Inc. (Thousand Oaks, CA, USA), GlaxoSmithKline plc, Ionis Pharmaceuticals (Carlsbad, CA, USA), Eli Lilly and Company, OliPass (Gyeonggi, Republic of Korea), Regeneron Pharmaceuticals (Tarrytown, NY, USA) and Theranexus (Lyon, France) (2020-21). David L Bennett has received research funding from Eli Lilly and Company and AstraZeneca, and has received an industrial partnership grant from the Biotechnology and Biological Sciences Research Council (BBSRC) (Swindon, UK) and AstraZeneca. David L Bennett reports grants and contracts for a number of studies from the following: the UK Research and Innovation (Swindon, UK) (Versus MR/ W002388/1), Medical Research Council (MRC) (MR/T020113/1), BBSRC (BB/S006788/1), Action Medical Research for Children (West Sussex, UK), MRC Research Grant, Wellcome Trust Senior Clinical Scientist Fellowship, Novo Nordisk Foundation (Hellerup, Denmark), European Union Horizon 2020, MRC Clinical Research Training Fellowship and Wellcome Trust Strategic Award. Dinesh Selvarajah reports membership of the advisory boards of Impeto Medical (Issy-les-Moulineaux, France) (2017), PelliTec Inc. (Chester, UK) (2020) and FeetMe Inc. (Paris, France) (2019). Cindy Cooper reports membership of the following committees: the National Institute for Health and Care Research (NIHR) Clinical Trial Unit (CTU) Support Funding Committee (2016 to present), NIHR CTU Standing Advisory Committee (2016-22), NIHR Programme Grant for Applied Research Subcommittee (2017-21) and Trial Steering Committees for other NIHR-funded trials.

^{*}Corresponding author solomon.tesfaye@nhs.net

Published October 2022 DOI: 10.3310/RXUO6757

Plain English summary

The OPTION-DM RCT

Health Technology Assessment 2022; Vol. 26: No. 39

DOI: 10.3310/RXUO6757

NIHR Journals Library www.journalslibrary.nihr.ac.uk

Plain English summary

The number of people with diabetes is growing rapidly in the UK and is predicted to rise to over 5 million by 2025. Diabetes causes nerve damage that can lead to severe painful symptoms in the feet, legs and hands. One-quarter of all people with diabetes experience these symptoms, known as 'painful diabetic neuropathy'. Current individual medications provide only partial benefit, and in only around half of patients. The individual drugs, and their combinations, have not been compared directly against each other to see which is best.

We conducted a study to see which treatment pathway would be best for patients with painful diabetic neuropathy. The study included three treatment pathways using combinations of amitriptyline, duloxetine and pregabalin. Patients received all three treatment pathways (i.e. amitriptyline treatment for 6 weeks and pregabalin added if needed for a further 10 weeks, duloxetine treatment for 6 weeks and pregabalin added if needed for a further 10 weeks and pregabalin treatment for 6 weeks and amitriptyline added if needed for a further 10 weeks); however, the order of the treatment pathways was decided at random. We compared the level of pain that participants experienced in each treatment pathway to see which worked best.

On average, people said that their pain was similar after each of the three treatments and their combinations. However, two treatments in combination helped some patients with additional pain relief if they only partially responded to one. People also reported improved quality of life and sleep with the treatments, but these were similar for all the treatments. In the health economic analysis, the value for money and quality of life were similar for each pathway, and this resulted in uncertainty in the cost-effectiveness conclusions, with no one pathway being more cost-effective than the others. The treatments had different side effects, however; pregabalin appeared to make more people feel dizzy, duloxetine made more people nauseous and amitriptyline resulted in more people having a dry mouth. The pregabalin supplemented by amitriptyline pathway had the smallest number of treatment discontinuations due to side effects and may be the safest for patients.

Health Technology Assessment

ISSN 1366-5278 (Print)

ISSN 2046-4924 (Online)

Impact factor: 4.014

Launched in 1997, *Health Technology Assessment* (HTA) has an impact factor of 4.014 and is ranked 27th (out of 108 titles) in the 'Health Care Sciences & Services' category of the Clarivate 2021 Journal Citation Reports (Science Edition). It is also indexed by MEDLINE, CINAHL (EBSCO Information Services, Ipswich, MA, USA), Embase (Elsevier, Amsterdam, the Netherlands), NCBI Bookshelf, DOAJ, Europe PMC, the Cochrane Library (John Wiley & Sons, Inc., Hoboken, NJ, USA), INAHTA, the British Nursing Index (ProQuest LLC, Ann Arbor, MI, USA), Ulrichsweb™ (ProQuest LLC, Ann Arbor, MI, USA) and the Science Citation Index Expanded™ (Clarivate™, Philadelphia, PA, USA).

This journal is a member of and subscribes to the principles of the Committee on Publication Ethics (COPE) (www.publicationethics.org/).

Editorial contact: journals.library@nihr.ac.uk

The full HTA archive is freely available to view online at www.journalslibrary.nihr.ac.uk/hta.

Criteria for inclusion in the Health Technology Assessment journal

Reports are published in *Health Technology Assessment* (HTA) if (1) they have resulted from work for the HTA programme, and (2) they are of a sufficiently high scientific quality as assessed by the reviewers and editors.

Reviews in *Health Technology Assessment* are termed 'systematic' when the account of the search appraisal and synthesis methods (to minimise biases and random errors) would, in theory, permit the replication of the review by others.

HTA programme

Health Technology Assessment (HTA) research is undertaken where some evidence already exists to show that a technology can be effective and this needs to be compared to the current standard intervention to see which works best. Research can evaluate any intervention used in the treatment, prevention or diagnosis of disease, provided the study outcomes lead to findings that have the potential to be of direct benefit to NHS patients. Technologies in this context mean any method used to promote health; prevent and treat disease; and improve rehabilitation or long-term care. They are not confined to new drugs and include any intervention used in the treatment, prevention or diagnosis of disease.

The journal is indexed in NHS Evidence via its abstracts included in MEDLINE and its Technology Assessment Reports inform National Institute for Health and Care Excellence (NICE) guidance. HTA research is also an important source of evidence for National Screening Committee (NSC) policy decisions.

This report

The research reported in this issue of the journal was funded by the HTA programme as project number 15/35/03. The contractual start date was in June 2016. The draft report began editorial review in September 2021 and was accepted for publication in March 2022. The authors have been wholly responsible for all data collection, analysis and interpretation, and for writing up their work. The HTA editors and publisher have tried to ensure the accuracy of the authors' report and would like to thank the reviewers for their constructive comments on the draft document. However, they do not accept liability for damages or losses arising from material published in this report.

This report presents independent research funded by the National Institute for Health and Care Research (NIHR). The views and opinions expressed by authors in this publication are those of the authors and do not necessarily reflect those of the NHS, the NIHR, the HTA programme or the Department of Health and Social Care. If there are verbatim quotations included in this publication the views and opinions expressed by the interviewees are those of the interviewees and do not necessarily reflect those of the authors, those of the NHS, the NIHR, the HTA programme or the Department of Health and Social Care.

Copyright © 2022 Tesfaye *et al.* This work was produced by Tesfaye *et al.* under the terms of a commissioning contract issued by the Secretary of State for Health and Social Care. This is an Open Access publication distributed under the terms of the Creative Commons Attribution CC BY 4.0 licence, which permits unrestricted use, distribution, reproduction and adaption in any medium and for any purpose provided that it is properly attributed. See: https://creativecommons.org/licenses/by/4.0/. For attribution the title, original author(s), the publication source – NIHR Journals Library, and the DOI of the publication must be cited.

Published by the NIHR Journals Library (www.journalslibrary.nihr.ac.uk), produced by Prepress Projects Ltd, Perth, Scotland (www.prepress-projects.co.uk).

NIHR Journals Library Editor-in-Chief

Professor Ken Stein Professor of Public Health, University of Exeter Medical School, UK

NIHR Journals Library Editors

Professor John Powell Consultant Clinical Adviser, National Institute for Health and Care Excellence (NICE), UK, and Professor of Digital Health Care, Nuffield Department of Primary Care Health Sciences, University of Oxford, UK

Professor Andrée Le May Chair of NIHR Journals Library Editorial Group (HSDR, PGfAR, PHR journals) and Editor-in-Chief of HSDR, PGfAR, PHR journals

Professor Matthias Beck Professor of Management, Cork University Business School, Department of Management and Marketing, University College Cork, Ireland

Dr Tessa Crilly Director, Crystal Blue Consulting Ltd, UK

Dr Eugenia Cronin Consultant in Public Health, Delta Public Health Consulting Ltd, UK

Dr Peter Davidson Interim Chair of HTA and EME Editorial Board. Consultant Advisor, School of Healthcare Enterprise and Innovation, University of Southampton, UK

Ms Tara Lamont Senior Adviser, School of Healthcare Enterprise and Innovation, University of Southampton, UK

Dr Catriona McDaid Reader in Trials, Department of Health Sciences, University of York, UK

Professor William McGuire Professor of Child Health, Hull York Medical School, University of York, UK

Professor Geoffrey Meads Emeritus Professor of Wellbeing Research, University of Winchester, UK

Professor James Raftery Professor of Health Technology Assessment, School of Healthcare Enterprise and Innovation, University of Southampton, UK

Dr Rob Riemsma Consultant Advisor, School of Healthcare Enterprise and Innovation, University of Southampton, UK

Professor Helen Roberts Professor of Child Health Research, Child and Adolescent Mental Health, Palliative Care and Paediatrics Unit, Population Policy and Practice Programme, UCL Great Ormond Street Institute of Child Health, London, UK

Professor Jonathan Ross Professor of Sexual Health and HIV, University Hospital Birmingham, UK

Professor Helen Snooks Professor of Health Services Research, Institute of Life Science, College of Medicine, Swansea University, UK

Professor Ken Stein Professor of Public Health, University of Exeter Medical School, UK

Professor Jim Thornton Professor of Obstetrics and Gynaecology, Faculty of Medicine and Health Sciences, University of Nottingham, UK

Please visit the website for a list of editors: www.journalslibrary.nihr.ac.uk/about/editors

Editorial contact: journals.library@nihr.ac.uk